Author's response to reviews

Title: The use of superparamagnetic iron oxide-polymer composite microcapsules for magnetic resonance-guided high-intensity focused ultrasound cancer surgery

Authors:

Yang Sun (sy19850905@126.com)
Yuanyi Zheng (zhengyuanyi@gmail.com)
Pan Li (cqlipan@163.com)
Chengcheng Niu (cici204675@163.com)
Yuping Gong (gongyuping1982@163.com)
Rongzhong Huang (rzhuang@live.com)
Zhibiao Wang (wangzhibiao@haifu.com.cn)
Zhigang Wang (wzg62942443@163.com)
Haitao Ran (ranhaitao66@gmail.com)

Version: 4 Date: 21 August 2014

Author's response to reviews: see over
Dear Editor(s):

This is a revised version of manuscript entitled “The Use of Superparamagnetic Iron Oxide-Polymer Composite Microcapsules for Magnetic Resonance-Guided High-Intensity Focused Ultrasound Cancer Surgery.”

We appreciate the editors’ comments, “if you look at lines 125 and 308 I believe that the authors are discussing temperature but the units are designated by a box instead of something temperature related. The same thing happened on 189-192 when the authors designated 3 separate groups but the groups are each designated with a box instead of another identifier”.

The units in lines 125 and 308 is “□” as shown in the following pictures, we do not know why they are showing a box in the manuscript, we have changed these units in the manuscript.

![Picture of line 125 with corrected unit]

Also, we have changed the group number in the revised manuscript as shown in the following picture.

![Picture of line 189 with corrected group information]

Regards,
Haitao Ran